Author's response to reviews

Title: Nutrition promotion approaches preferred by Australian adolescents attending schools in disadvantaged neighbourhoods: a qualitative study

Authors:

Lena D Stephens (l.stephens@deakin.edu.au)
Sarah A McNaughton (sarah.mcnaughton@deakin.edu.au)
David Crawford (david.crawford@deakin.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)

Version: 8  Date: 3 May 2015

Author's response to reviews:

Dear Editor,

Editor's comment and authors' responses for Manuscript No. 1524195445142811: "Nutrition promotion approaches preferred by Australian adolescents attending schools in disadvantaged neighbourhoods: a qualitative study."

Thank you for providing further comments on our paper, which were greatly appreciated. Our response is detailed below.

1. Please review the abstract on the submission system as it currently contains duplications.
   - As requested, the abstract has been amended accordingly.

2. Title Page:

   Please include the email addresses of all authors on the title page. Please ensure that it is identical to the one entered on the submission system.

   - The required email addresses have been supplied on the title page of the manuscript.

Thank you for considering our manuscript for publication in BMC Pediatrics.

Yours sincerely,

Lena Stephens, PhD
Mail: Centre for Physical Activity and Nutrition Research, School of Exercise and Nutrition Sciences, Deakin University, 221 Burwood Highway, Burwood VIC 3125, AUSTRALIA

Email: l.stephens@deakin.edu.au
Phone: +61-3-9244-6328
Fax: +61-3-9244-6017